

CRISPR editing of sftb-1/SF3B1 in Caenorhabditis elegans allows the identification of synthetic interactions with cancer-related mutations and the chemical inhibition of splicing

Xènia Serrat, Dmytro Kukhtar, Eric Cornes, Anna Esteve-Codina, Helena Benlloch, Germano Cecere, Julián Cerón

▶ To cite this version:

Xènia Serrat, Dmytro Kukhtar, Eric Cornes, Anna Esteve-Codina, Helena Benlloch, et al.. CRISPR editing of sftb-1/SF3B1 in Caenorhabditis elegans allows the identification of synthetic interactions with cancer-related mutations and the chemical inhibition of splicing. PLoS Genetics, 2019, 15 (10), pp.e1008464. 10.1371/journal.pgen.1008464. pasteur-03258502

HAL Id: pasteur-03258502 https://pasteur.hal.science/pasteur-03258502v1

Submitted on 11 Jun2021

HAL is a multi-disciplinary open access archive for the deposit and dissemination of scientific research documents, whether they are published or not. The documents may come from teaching and research institutions in France or abroad, or from public or private research centers. L'archive ouverte pluridisciplinaire **HAL**, est destinée au dépôt et à la diffusion de documents scientifiques de niveau recherche, publiés ou non, émanant des établissements d'enseignement et de recherche français ou étrangers, des laboratoires publics ou privés.



Distributed under a Creative Commons Attribution 4.0 International License



G OPEN ACCESS

Citation: Serrat X, Kukhtar D, Cornes E, Esteve-Codina A, Benlloch H, Cecere G, et al. (2019) CRISPR editing of *sttb-1/SF3B1* in *Caenorhabditis elegans* allows the identification of synthetic interactions with cancer-related mutations and the chemical inhibition of splicing. PLoS Genet 15(10): e1008464. https://doi.org/10.1371/journal. pgen.1008464

Editor: H. Leighton Grimes, Cincinnati Children's Hospital Medical Center, UNITED STATES

Received: July 12, 2019

Accepted: October 5, 2019

Published: October 21, 2019

Peer Review History: PLOS recognizes the benefits of transparency in the peer review process; therefore, we enable the publication of all of the content of peer review and author responses alongside final, published articles. The editorial history of this article is available here: https://doi.org/10.1371/journal.pgen.1008464

Copyright: © 2019 Serrat et al. This is an open access article distributed under the terms of the Creative Commons Attribution License, which permits unrestricted use, distribution, and reproduction in any medium, provided the original author and source are credited.

Data Availability Statement: The RNA sequencing data generated in this study have been deposited in

RESEARCH ARTICLE

CRISPR editing of *sftb-1/SF3B1* in *Caenorhabditis elegans* allows the identification of synthetic interactions with cancer-related mutations and the chemical inhibition of splicing

Xènia Serrat¹, Dmytro Kukhtar¹, Eric Cornes², Anna Esteve-Codina^{3,4}, Helena Benlloch¹, Germano Cecere², Julián Cerón¹*

 Modeling human diseases in *C. elegans* Group, Genes, Disease and Therapy Program, Institut d'Investigació Biomèdica de Bellvitge–IDIBELL, Barcelona, Spain, 2 Mechanisms of Epigenetic Inheritance, Department of Developmental and Stem Cell Biology, Institut Pasteur, UMR3738, CNRS, Paris, France, 3 CNAG-CRG, Centre for Genomic Regulation (CRG), The Barcelona Institute of Science and Technology, Barcelona, Spain, 4 Universitat Pompeu Fabra (UPF), Barcelona, Spain

* jceron@idibell.cat

Abstract

SF3B1 is the most frequently mutated splicing factor in cancer. Mutations in *SF3B1* likely confer clonal advantages to cancer cells but they may also confer vulnerabilities that can be therapeutically targeted. *SF3B1* cancer mutations can be maintained in homozygosis in *C*. *elegans*, allowing synthetic lethal screens with a homogeneous population of animals. These mutations cause alternative splicing (AS) defects in *C. elegans*, as it occurs in *SF3B1*-mutated human cells. In a screen, we identified RNAi of U2 snRNP components that cause synthetic lethality with *sftb-1/SF3B1* mutations. We also detected synthetic interactions between *sftb-1* mutants and cancer-related mutations in *uaf-2/U2AF1* or *rsp-4/SRSF2*, demonstrating that this model can identify interactions between mutations that are mutually exclusive in human tumors. Finally, we have edited an SFTB-1 domain to sensitize *C. elegans* to the splicing modulators pladienolide B and herboxidiene. Thus, we have established a multicellular model for *SF3B1* mutations amenable for high-throughput genetic and chemical screens.

Author summary

SF3B1 encodes a splicing factor frequently mutated in hematological malignancies, and with less frequency in other solid tumors. Each *SF3B1* mutation is predominant in a cancer type, indicating cell type-specific pathogenic mechanisms. We have reproduced three cancer-related *SF3B1* missense mutations in the *C. elegans* ortholog *sftb-1* and they do not cause any overt phenotypes, but few alterations in splicing. However, the combination of these mutations has additive effects, hampering SFTB-1 functions and allowing the study of synthetic interactions that may result in a defective spliceosome structure. As a result of

PLOS GENETICS

the Gene Expression Omnibus repository and are available under the accession number GEO: GSE129642. All other relevant data are within the manuscript and its Supporting Information files.

Funding: This work has been supported by a grant from the Instituto de Salud Carlos III (ISCIII), http:// www.isciii.es/, to JC (PI15-00895), co-funded by FEDER funds/European Regional Development Fund (ERDF) — a way to build Europe, https://ec. europa.eu/regional_policy/en/funding/erdf/. AEC is funded by ISCIII, http://www.isciii.es/, of the MINECO (reference PT17/0009/0019) and cofinanced by FEDER, https://ec.europa.eu/regional_ policy/en/funding/erdf/. EC is supported by a Pasteur-Roux postdoctoral fellowship program, https://research.pasteur.fr/en/call/call-forapplication-pasteur-roux-and-pasteurcantarinipostdoctoral-fellowships-2019-spring-session/. XS has an FPU PhD fellowship from MINECO, http:// www.educacionyfp.gob.es/servicios-al-ciudadanomecd/becas-ayudas.html. DK has an FI AGAUR fellowship from Generalitat de Catalunya, http:// agaur.gencat.cat/en/inici/. The funders had no role in study design, data collection and analysis, decision to publish, or preparation of the manuscript.

Competing interests: The authors have declared that no competing interests exist.

an RNAi screen, we identified other proteins of the U2 snRNP complex as targets for synthetic lethality. The *sftb-1* mutant strains also allow the identification of mutations in other splicing factors that are mutually exclusive with *SF3B1* mutations in tumors. *SF3B1* cancer mutations are heterozygous, and interestingly, the most promising therapeutic venue is based on pladienolide B (PB) derivatives that hit the wild-type and the mutated protein. We have edited SFTB-1 to sensitize it to pladienolide B and herboxidiene (HB), therefore making *C. elegans* suitable for assaying the toxicity and efficacy of their chemical derivatives. Thus, our model represents a pre-clinical system for investigating both synthetic lethality and chemotherapies. Besides that, we have generated the first *C. elegans* strain in which splicing can be chemically modulated by PB and HB.

Introduction

Deregulated RNA splicing is emerging as a new hallmark of cancer following the discovery of several splicing factor genes harboring somatic mutations in different tumor types [1]. *SF3B1*, a core component of the U2 snRNP, is the most frequently mutated splicing factor in human cancers. Somatic heterozygous mutations in *SF3B1* are particularly prevalent in myelodysplastic syndromes (MDS)—up to 80% in refractory anemia with ring sideroblasts [2–4]; and in 15% of chronic lymphocytic leukemia (CLL) [5–7]. *SF3B1* mutations have also been reported in solid tumors, including 20% of uveal melanomas (UM) [8–10], 3% of pancreatic ductal adenocarcinomas [11], and 1.8% of breast cancers [12]. RNA-sequencing (RNA-seq) analyses in different tumor types and cell lines with *SF3B1* missense mutations, including the most prevalent substitution K700E, identified distinct alternative splicing (AS) defects [13–15].

Recently, CRISPR/Cas9 genome editing has allowed researchers to faithfully reproduce human pathological mutations in animal models. Human SF3B1 mutations have been introduced in cell lines [15-18], but mimicking these mutations in multicellular organisms has only been achieved by conditional alleles in murine models [19]. Taking advantage of the extraordinary conservation of splicing factors across evolution and the ease of genetic manipulation of *C. elegans*, we established a multicellular model to study *SF3B1* cancer-related mutations. Spliceosome components, and particularly SF3B1, are being intensively studied as targets of antitumor drugs [20–22]. Some natural compounds modulate splicing by targeting SF3B1. Among them, pladienolide B is particularly relevant because H3B-8800, a PB derivative, has antitumoral effects and is in clinical trials [23]. The modulatory activity of PB is highly dependent on SF3B1 structure, and single amino acid substitutions cause resistance to PB [24,25]. Hence, particular residues in the drug binding site that are not conserved in SFTB-1 may confer resistance to PB in worms, as it has been recently observed in yeast [26]. We edited four amino acids to humanize the HEAT repeat 15 in SFTB-1, making C. elegans sensitive to splicing modulators. Thus, we have created the first C. elegans strain sensitive to PB and HB, which would facilitate the chemical modulation of splicing in this model system.

Altogether, the multicellular model of *SF3B1* mutations described in this study provides a new pre-clinical platform for identifying new targets and small molecules for use in cancer therapies.

Results

sftb-1, the *C. elegans* ortholog of *SF3B1*, is a ubiquitously expressed gene essential for development

The *C. elegans* protein SFTB-1 is 66% identical to human SF3B1 in terms of amino acid composition. Homology is particularly high at the HEAT domain, reaching 89% identity, and the

most frequently mutated amino acids in cancer are conserved. The SFTB-1 sequence also conserves some of the U2AF ligand motifs (ULMs) that bind U2AF homology motifs (UHMs) present in other splicing factors (Figs <u>1A</u> and <u>S1</u>) [27].

As expected for a core splicing factor, the CRISPR-engineered endogenous fluorescent reporter mCherry::SFTB-1 showed ubiquitous expression in somatic and germ cells, being absent in mature sperm only (Fig 1B and 1C). We also used CRISPR to generate the *sftb-1* (*cer6*) mutation, a deletion allele that produces a premature stop codon (S2 Fig) and causes an arrest at early larval stages, confirming that *sftb-1* is essential for development (Fig 1D and 1E). *sftb-1(cer6)* homozygous animals completed embryonic development but arrested as larvae that could undergo few germ cell divisions (S2 Fig). Since embryos depleted of *sftb-1* transcripts are not viable [28], our observation indicates that maternal wild-type (WT) product can still be present in the early larva. We formally tested this hypothesis by combining the *sftb-1* deletion allele and the *sftb-1* endogenous mCherry reporter in the same strain (S2 Fig). The compound heterozygotes produced a normal brood size and segregated 19.39% non-red *cer6* homozygous mothers, we observed a small percentage of early L1 larvae with dim red fluorescent expression that later became non-red and developmentally arrested larvae (S2 Fig), indicating that there is some SFTB-1 maternal product reaching the early L1 stage.

Different sftb-1 missense mutations display additive effects when combined

We edited the *C. elegans* genome to mimic the K700E mutation, the most frequent *SF3B1* substitution in tumors. *sftb-1*[K718E] homozygous animals did not display any obvious phenotypes. Thus, we decided to reproduce two other *SF3B1* cancer-related mutations. R625C is a particularly prevalent mutation in uveal melanoma, and Q534P is less common but is predicted to have a strong impact on SF3B1 structure due to its location in an α -helix [29]. Similar to *sftb-1*[K718E], the two additional missense mutations, *sftb-1*[R643C] and *sftb-1*[Q552P] (S3 Fig), did not cause any evident phenotypes in *C. elegans*.

We were interested in producing animals with defective *sftb-1* function resulting in a viable and traceable phenotype for investigating modifiers of altered SFTB-1 activity. Since the effects of MDS-related missense mutations were reported to be additive in the yeast *SF3B1* ortholog Hsh155 [30], we generated three strains with each of the possible pairs of mutations: *sftb-1* [R643C, K718E], *sftb-1*[Q552P, K718E], and *sftb-1*[Q552P, R643C]. None of the three double mutants showed an obvious phenotype. However, animals simultaneously carrying the three missense mutations *sftb-1*[Q552P, R643C, K718E] displayed several temperature-sensitive developmental defects including partial sterility, protruding vulva, and developmental delay (Figs <u>1F-1H</u> and <u>S3</u>). These results suggest that each of the three missense mutations are combined.

Distinct *sftb-1* alleles produce alterations in alternative splicing and gene expression

The most commonly described AS defect caused by cancer-associated *SF3B1* mutations is the incorrect recognition of 3' splice sites, producing aberrant transcripts [13,14,18,31,32]. We investigated in *C. elegans* the transcriptional consequences of the K718E mutation, an equivalent of the most frequent *SF3B1* mutation K700E, by RNA-seq of two different strains with two independent K718E alleles (*cer3* and *cer7*) and their WT siblings at the L4 stage. Then, we used rMATS for robust detection of differential AS events between mutant and WT worms [33]. Using both splice junction and exon body reads, our analysis detected 21,985 AS events, of which 134 were significantly altered (FDR < 0.05) (S1 Table). A more restricted cut-off

PLOS GENETICS



Fig 1. Characterization of *sftb-1*, the *C. elegans* **ortholog of** *SF3B1*, and different *sftb-1* **mutations**. (A) Number of *SF3B1* missense mutations annotated in COSMIC database [75] at each amino acid position, with labels placed on the relevant residues for this study. A schematic structure of the SF3B1 protein is shown below, where purple, yellow, and blue boxes represent the U2AF ligand motifs (ULMs), the p14-interacting region, and the 20 HEAT repeats, respectively. The bars at the bottom line indicate residues that are conserved in the *C. elegans* protein. (B) Representative image of an embryo, L2, and L4 larvae with mCherry::SFTB-1 endogenous expression. The fluorescence image (top panel) is merged with the corresponding differential interference contrast (DIC) image (bottom panel). (C) Fluorescence images of dissected gonads from adult worms expressing mCherry::SFTB-1. Gonads were stained with DAPI to visualize nuclei. Insets show differentiated sperm nuclei, where mCherry::SFTB-1 is not expressed. (D) Left: diagram depicting how the *cer6* deletion allele is maintained in heterozygosis. The WT *sftb-1* copy is genetically linked to a red fluorescent marker (*axTi615*). Right: worm body length of WT and heterozygotes (red, n = 475), and *cer6* homozygotes (gray, n = 134) grown for 48 h at 20° C (N = 3). (E) Represent percentages observed in each replicate (n = 119, 116; N = 2). (G) Brood size of WT or *sftb-1* [Q552P, R643C, K718E] phenotypes at 25° C. Dots represent percentages observed in each replicate (n = 119, 116; N = 2). (G) Brood size of WT or *sftb-1* [Q552P, R643C, K718E] fertile worms at 25° C (n = 22, 35; N = 2). (H) *sftb-1*[Q552P, R643C, K718E] animals display developmental delay at 25° C. The proportion of the population at each stage is represented by the size of the dots (n = 119, 116; N = 2). In (D) and (G), dots represent measures in individual worms, and overlaid Tukey-style boxplots represent the median and interquartile range (IQR). Whiskers extend to the minimum and maxim

https://doi.org/10.1371/journal.pgen.1008464.g001

(inclusion level difference > 0.1), similar to that used in a previous rMATS analysis of *SF3B1*mutated samples [32], reduced the list of significant AS events to 78, with skipped exon (SE) being the most frequent event (Fig 2A and 2B).

Then, we investigated whether the two *sftb-1* alleles associated with a visible phenotype (the triple mutation allele *cer39*, and the deletion allele *cer6*) primarily induced cryptic 3' splice site selection. It is worth noting that, in this experiment, larvae were harvested and processed for RNA isolation at the L2 stage to avoid possible indirect effects due to developmental defects observed after this stage. Both alleles were associated with a high number of deregulated AS events as determined by rMATS, and the predominant alteration was also exon skipping (Fig 2B).

We then explored if there were differentially expressed protein-coding (Fig 2C) and noncoding (Fig 2D) transcripts in the three RNA-seq datasets (S2 Table). As expected, the highest number of significantly altered transcripts was observed in the *cer6* allele.

A functional enrichment analysis uncovered biological pathways that were significantly affected by the distinct *sftb-1* alleles (S4 Fig). The K718E mutation caused the downregulation of transcripts related to mitochondrial functions, which were also downregulated by the deletion allele *cer6*. Both *cer39* and *cer6* alleles induced the upregulation of transcripts involved in mRNA processing, and the downregulation of different metabolic pathways. Thus, it seems that the common reaction to an impaired splicing is to increase the splicing components and reduce the metabolism.

Interestingly, there is a partial overlap between the transcripts deregulated by the triple mutation allele *cer39* and the deletion allele *cer6* (Fig 2E). We also found few AS events that are similarly altered in distinct *sftb-1* mutants. The largest overlap was again observed between the *cer39* and the *cer6* alleles, which shared 80 deregulated events (S5 Fig). Thus, the additive effect of three missense mutations produces a similar impact on gene expression as the deletion allele but with a considerable number of unique AS changes.

In summary, we identified gene expression and AS alterations that are more frequent as *sftb-1* function becomes progressively compromised, with SE being the predominant AS defect.

A subset of U2 snRNP and U2-associated components synthetically interacts with *sftb-1* mutants

In the search for selective vulnerabilities in *sftb-1* defective backgrounds, we hypothesized that *sftb-1* mutant animals would be more sensitive to knockdown of other splicing factors than their WT counterparts. Based on a previous *C. elegans* RNA interference (RNAi) collection of



Fig 2. Different *sftb-1* **mutations induce alternative splicing and gene expression alterations.** (A) Types of AS events. Gray boxes represent constitutive exons, while colored boxes represent alternatively spliced exons. (B) Total number of significant (FDR<0.05) AS alterations detected by rMATS with an inclusion level difference > 0.1. Event types are color coded as in panel A. Light colors represent AS events that were significantly more included in WT samples, while dark colors symbolize splicing alterations associated to mutant samples. Inset doughnut charts show the distribution of significant events by event type. In the three mutants, skipped exon (SE) was the most predominant alteration. (C) MA-plots displaying changes in protein-coding transcript (padj<0.05) are colored in blue. (D) MA-plots of expression changes in non-coding transcripts, represented as in panel C. Transcripts that are using directed are colored by transcript type. (E) Venn diagrams showing the overlap between transcripts that are upregulated (left) or downregulated (right) in the three datasets. The number of transcripts at the intersections is shown, as well as the total number of significantly up- or downregulated transcripts in each mutant strain (padj<0.05).

https://doi.org/10.1371/journal.pgen.1008464.g002

splicing factors [34], we compiled and assayed a total of 104 RNAi clones targeting proteins that act in different steps of the splicing reaction. This led us to identify 27 splicing factors whose knockdown phenotype was enhanced in at least one of the three *sftb-1* single mutants, suggesting possible synthetic interactions. Among the potential candidates, we identified proteins present in pre-catalytic and catalytic spliceosomes, as well as few auxiliary factors, indicating that *sftb-1* missense mutations might impact other steps of the splicing reaction besides early spliceosome assembly (S3 Table).

We selected three U2 snRNP components, including *sftb-1*, and two U2-associated proteins for further validation based on their presence in the U2 snRNP and the phenotypes observed (Fig 3A). The identification of *sftb-1* as a candidate certified our approach since it demonstrated that any RNAi affecting *sftb-1*-related functions would have a stronger effect on *sftb-1* mutants. Since a partial loss-of-function mutant for the candidate *teg-4* was available, we proceeded to validate the interaction between *sftb-1* mutations and *teg-4*, the *C. elegans* ortholog of *SF3B3*, which physically interacts with SF3B1 HEAT repeats 3, 5, and 6 [29]. We found that the K718E mutation exacerbated the phenotype of *teg-4(oz210)* at 20°C [35], whereas the Q552P mutation did so to a lesser extent (Fig 3B).

Then, we validated the rest of the candidates using RNAi. As we had observed that *sftb-1* mutations had an additive effect (Fig 1F–1H), we reasoned that *sftb-1* double mutants might be more sensitive to splicing perturbations than single mutants. Consistently, *sftb-1(RNAi)* led to an additive loss of function in double mutants, which arrested at earlier stages compared to WT worms (Fig 3C). A similar result was obtained with knockdown of *uaf-2/U2AF1* (Fig 3D), another candidate from the RNAi screen.

Among the remaining U2 and U2-associated candidates, we selected *mog-2/SNRPA1* and *smr-1/SMNDC1* for further validation in the double-mutation backgrounds. Strikingly, whereas RNAi of *mog-2* resulted in very low penetrant sterility in WT worms, the incidence of the phenotype incremented to over 70% in the three *sftb-1* double-mutation strains (Fig 3E). A similar synthetic phenotype was observed upon knockdown of *smr-1* (Fig 3F). Finally, *sftb-1* double mutants were more sensitive to *teg-4(RNAi)* (Fig 3G). This former synthetic interaction was not as strong as the one observed with *mog-2(RNAi)* and *smr-1(RNAi)* but was supported by the interaction between genetic alleles (Fig 3B).

We also found that heterozygous animals carrying the deletion allele *sftb-1(cer6)* were sensitive to RNAi of the five interactors (S6 Fig). Still, double mutant strains are better than *cer6* heterozygous animals for identifying genetic interactions (Figs 3 and S6).

Together, our results suggest that targeting other splicing factors such as *SF3B3* to compromise their function can be an opportunity to selectively impair cancer cells harboring *SF3B1* mutations.

The *C. elegans* mutations equivalent to human *U2AF1*^{S34P} and *SRSF2*^{P95H} genetically interact with *sftb-1*[Q552P, R643C, K718E] mutants

Along with *SF3B1* mutations, recurrent missense mutations in U2 small nuclear RNA auxiliary factor 1 (*U2AF1*) and serine/arginine-rich splicing factor 2 (*SRSF2*) were discovered in hematological malignancies [2], and later reported to be present in some solid tumors [36,37]. Somatic hotspot mutations in these three splicing factors are only found in heterozygosis and are mutually exclusive, suggesting redundant functional impact or limited tolerance to splicing perturbations.

Missense mutations in *U2AF1* are commonly found at position S34, which corresponds to a residue located within the two zinc finger domains through which U2AF1 interacts with the pre-mRNA [38]. Similarly, *SRSF2* is typically mutated at position P95, mainly bearing missense mutations but also in-frame insertions or deletions [2].



Fig 3. Synthetic interaction between *sftb-1* mutant strains and knockdown of U2 snRNP and U2 snRNP-related components. (A) Schematics of the RNAi screen. In brief, WT or mutant worms bearing any of the three *sftb-1* single mutations were fed with RNAi clones against a total of 104 splicing factors. 77 clones had the same effect in all genetic backgrounds. In contrast, we identified 27 clones that caused a stronger phenotype in at least one of the three single mutant strains compared to WT, which was indicative of a synthetic interaction. Listed are the 3 U2 snRNP and 2 U2 snRNP-related candidates from the screen that were selected for further validation. (B) *sftb-1*[K718E] and *sftb-1*[Q552P] synthetically interact with *teg-4(oz210)*. The number of hatched larvae (orange) and dead embryos (gray) laid by worms of the indicated genotype at 20°C is represented (n \geq 28; N = 3). *m*+*z*-, homozygous mutant progeny of heterozygous mothers. No significant differences were observed in the number of hatched larvae laid by *sftb-1*[K718E], *sftb-1*[R643C], and *sftb-1*[Q552P] compared to WT. (C) Worm length after 48 h of RNAi treatment against *gfp* (black) or *sftb-1* (blue) at 25°C (n \geq 150; N = 2). (D) Worm length after 48 h of RNAi treatment against *gfp* (black) or *sftb-1*(blue) at 25°C (n \geq 150; N = 2). In both (C) and (D), worm length was not significantly different between *sftb-1* mutants and WT upon *gfp(RNAi)*. (E-G) Mean percentage of sterile worms observed upon RNAi of *mog-2* (E; n \geq 30), *smr-1* (F; n \geq 27) and *teg-4* (G; n \geq 34) at 25°C.

Red dots indicate percent sterility observed in each replicate (N = 2). Gray bars, P₀-treated worms giving rise to <5 F₁ larvae and some dead embryos ('dead progeny' category); black bars, P₀-treated worms that laid neither larvae nor dead embryos ('no progeny' category). In (B), (C), and (D), data are shown as Tukey-style boxplots and overlaid dots representing measures in individual animals. Statistics: (B), (C), and (D), Kruskal-Wallis test with Dunn's multiple comparison test. Data were compared to *teg-4(oz210)* (B), or to WT animals fed with the same RNAi clone (C) and (D). (E-G), Fisher's exact test with Bonferroni correction for multiple comparison. * p<0.05, ** p<0.01, *** p<0.001.

https://doi.org/10.1371/journal.pgen.1008464.g003

Based on a recently reported synthetic interaction between $Srsf2^{P95H/+}$ and $Sf3b1^{K700E/+}$ in mice [39], and on the fact that *sftb-1* mutants were more sensitive to inactivation of some splicing factors (**Fig 3**), we wondered if the combination of *sftb-1*[K718E] with mutations equivalent to $U2AF1^{S34F}$ or $SRSF2^{P95H}$ would be tolerated in *C. elegans*.

Thus, we edited the worm orthologs of these genes to generate the *uaf*-2[S42F] and *rsp*-4 [P100H] mutant strains. Neither of these two mutations caused any obvious phenotypes nor dramatic changes in some AS events analyzed by RT-PCR (S7 Fig). In contrast to its murine counterpart $Srsf2^{P95}$, rsp-4[P100H] is tolerated in homozygosis [39].

We did not detect any interaction when combining each of these mutants with *sftb-1* [K718E] or the three *sftb-1* double-mutation strains. In contrast, *rsp-4*[P100H] and *uaf-2* [S42F] significantly increased the low percentage of sterile worms observed in the *sftb-1* [Q552P, R643C, K718E] background at 20°C (Fig 4A). The most dramatic effect was observed by the addition of the *uaf-2*[S42F] mutation to the *sftb-1* triple-mutation strain.

As a result of these synthetic interactions, we observed masculinization of the germline (Mog) and tumorous germline (Tum) phenotypes, which are hallmarks of germlines with altered splicing activity [34]. Thus, these synthetic interactions may cause a synergistic disruption of general splicing functions, perhaps through a flaw in spliceosomal architecture, although detailed information on the molecular basis of this interaction should be obtained by deep sequencing of mutant transcriptomes.

Moreover, this combination of *uaf-2* and *sftb-1* mutations induced developmental delay at 20°C (Fig 4B). These results show for the first time in an animal model a synthetic interaction between *sftb-1* and *uaf-2* missense mutations.

A humanized HEAT repeat 15 in SFTB-1 sensitizes *C. elegans* to pladienolide B and herboxidiene

The SF3B1 C-terminal region contains 20 structural motifs known as HEAT repeats, each one formed by two alpha helices joined by a short loop. These motifs display some structural plasticity and are involved in interactions with the pre-mRNA and other proteins [29]. Specific residues within SF3B1 HEAT repeats 15, 16, and 17 (H15-H17) interact with PB [40], and some amino acids in this region are not conserved in C. elegans (Fig 5A). In order to test if these sequence differences would affect the worms sensitivity to splicing modulators, we exposed WT animals and sftb-1 cancer-related mutants to PB and to the synthetic molecule sudemycin D6 [41], which also targets the SF3B complex. Both drugs had no visible effects when delivered in liquid medium or via microinjection (S4 Table). To reproduce the PB binding site in *sftb-1*, we used CRISPR to introduce four missense mutations (*cer144* allele) that mimic the human HEAT repeat 15 (Fig 5B). The resulting strain was apparently WT but became sensitive to PB, although the substitutions were not enough to overcome the resistance to sudemycin D6. sftb-1(cer144) mutant larvae exposed to 10, 50, and 100 μ M PB displayed a dose-dependent growth defect (Fig 5C). In addition, humanization of the HEAT repeat 15 also conferred sensitivity to herboxidiene (HB) (also known as GEX1A) (Fig 5D), which is predicted to be accommodated in the H15-H17 pocket similarly to PB [40,42].



Fig 4. Genetic interaction between *rsp-4/SRSF2*, *uaf-2/U2AF1*, and *sftb-1* mutant strains. (A) Mean percentage of sterile worms of the indicated genotype at 20°C (N = 2). Black dots denote observed values in each replicate. Statistical significance was determined with the Fisher's exact test with Bonferroni correction for multiple comparison. Lines indicate compared groups. ** p<0.01, **** p<0.0001. (B) *rsp-4*[P100H] and *uaf-2*[S42F] mutations gradually exacerbate the mild developmental delay displayed by the *sftb-1*[Q552P, R643C, K718E] mutant at 20°C. Dot sizes symbolize the percentage of worms at each developmental stage on any given day. In both panels, $n \ge 110$, N = 2.

https://doi.org/10.1371/journal.pgen.1008464.g004

Discussion

SF3B1 cancer-related mutations have been intensively studied in the past few years. They are particularly recurrent in MDS or UM, which commonly lack effective therapeutic approaches. In this study, we establish the first multicellular model for *SF3B1* mutations expressed non-conditionally from the endogenous locus. The rapid *C. elegans* life cycle (three days at 25°C) and the around 300 self-progeny obtained from a single hermaphrodite, among other features, make this model organism ideal for large-scale genetic and drug screens. The task of modeling human pathological mutations has been facilitated by the advent of the CRISPR/Cas9 genome editing technique, which has been efficiently implemented in this nematode [43,44].

As expected for a core splicing factor, and as previously reported in zebrafish [45] and mice [46], the *C. elegans SF3B1* ortholog is essential for development. Cancer cells bear *SF3B1* hotspot mutations in heterozygosis and depend on *SF3B1* WT function for viability [47]. However, homozygous *sftb-1* missense mutations are viable in *C. elegans*, although they generate defects that are unmasked when additional alterations in the spliceosome are introduced. Such a feature is advantageous for investigating vulnerabilities of *SF3B1*-mutated tumors, which are not only conferred by missense mutations, but also by the loss of one *SF3B1* WT copy [16,48].

Differently from MDS with *SF3B1* mutations, where aberrant transcripts have been directly related to phenotypic traits [32,49], transcripts affected by *sftb-1*[K718E] do not produce any obvious phenotypes. Still, *Sf3b1*^{K700E/+} murine models only partially recapitulate the hematopoietic phenotype characteristic of MDS [50,51] and CLL [52], although a modest overlap in

Α



0.5% 1% 0.5% 1% 50 μΜ 50 μΜ 1 μΜ 10 μΜ 50 μΜ 100 μΜ 1 μΜ 10 μΜ 50 μΜ 100 μΜ

Fig 5. Humanization of the SFTB-1 HEAT repeat 15 confers sensitivity to pladienolide B and herboxidiene. (A) Protein sequence alignment of residues in HEAT repeats 15 and 16. Amino acids that differ between worm SFTB-1 (*Ce*) and human SF3B1 (*Hs*) are colored in green. Positions of these residues in the human sequence are indicated. (B) CRISPR/Cas9 design to "humanize" the SFTB-1 HR15. In the wild-type allele, the crRNA sequence used is indicated by orange nucleotides, and the protospacer-adjacent motif (PAM) is underlined. In the mutant ("humanized") allele, synonymous mutations introduced to prevent partial recombination events and to improve primer specificity for mutant allele detection by PCR are indicated in blue. Missense mutations and the corresponding amino acids are colored in green. (C) Worm length after 48 h of drug treatment at 25°C ($n \ge 120$; N = 3). WT animals are represented as black dots, and *sftb-1(cer144)*

mutants as green dots. Doxycycline treatment induced larval arrest in both strains and was used as positive control. Data are shown as Tukey-style boxplots and overlaid dots represent measures in individual animals. The horizontal line at 1000 μ m indicates the expected body length of an adult worm. Body length of mutant worms treated with 50 μ M doxycycline and 10 μ M and 50 μ M PB differed statistically from that of mutant worms treated with 0.5% DMSO (p<0.0001). The same applied when comparing mutant worms treated with 100 μ M PB and 1% DMSO. Significance was determined with the Kruskal-Wallis test with Dunn's multiple comparison test. (D) Worm length after 48 h of herboxidiene or doxycycline (control) treatment at 25°C (n \geq 139; N = 3). Data are represented and analyzed as in panel C (n.s., no significant difference).

https://doi.org/10.1371/journal.pgen.1008464.g005

PLOS GENETICS

aberrant splicing events between $Sf3b1^{K700E/+}$ mice and MDS patients with SF3B1 mutations was detected [39].

RNA-seq of *sftb-1*[K718E] animals did not identify the aberrant use of alternative 3' splice sites (A3'SS) as the predominant splicing defect caused by this mutation, as has been observed in human cells [13,14,16,18,31,32,52,53]. Indeed, in three distinct *sftb-1* mutants, we found a mild effect on AS, with SE being the most abundant splicing defect. Such disparity in the consequences of the K700E mutation between humans and nematodes may be explained by different factors. First, diverse computational pipelines have been used to assess differences in alternative splicing between WT and SF3B1-mutated samples, resulting in distinct conclusions. For instance, Liberante and coworkers reanalyzed previously published data and found that SF3B1 mutation was mostly associated with differential skipped exon events, rather than A3'SS selection [15]. Second, intronic elements directing splicing have evolved differently in C. elegans compared to other organisms. C. elegans introns are typically shorter, do not have a consensus branch site (BS) sequence, and the vertebrate polypyrimidine (Py) tract is replaced by the consensus UUUUCAG/R sequence [54]. Third, only ~35% of genes are alternatively spliced in C. elegans [55] compared to 95% in humans [56]. Fourth, cell type-specific alternative isoforms generated by sftb-1 mutations could have been underrepresented in our analyses as we sequenced the transcriptome of whole animals, and A3'SS selection can be regulated in a tissue-specific manner in C. elegans [57]. And fifth, the nonsense-mediated decay (NMD) system could efficiently eliminate transcripts with A3'SS resulting from sftb-1 mutations, as occurs in human cells [13]. However, we did not detect predominance of downregulation among the differentially expressed transcripts (Fig 2C and 2D).

Beyond its action on A3'SS usage, *SF3B1* missense mutations could have an impact on the structure, the interaction with other proteins in the spliceosome, or the robustness of the protein. In fact, the PB derivative H3B-8800, which induces lethality in spliceosome-mutant cancers by modulating the SF3B complex, affects AS activity similarly in WT and in *SF3B1*^{K700E} human cells [23]. Therefore, *SF3B1*-mutated cells present vulnerabilities that can be exploited in synthetic lethal screens. We found that *sftb-1* mutants are sensitive to RNAi of different splicing factors, including itself, and we further validated these synthetic interactions with the inactivation of other components of the U2 snRNP. These results point to other components of the spliceosome as targets to preferentially kill cells with deficiencies in SF3B1 activity. Knowing the suitability of *sftb-1* mutants for RNAi screens, the collection of RNAi clones to be tested should be expanded, with particular interest placed on deubiquitinases and chromatin factors that have been related to SF3B1 functions [48,58].

Pladienolide and related compounds have been proven to be effective in selectively killing cells with *SF3B1* mutations [23,50,59]. We tested pladienolide B and sudemycin D6 in WT animals and *sftb-1* cancer-related mutants but did not observe any effects (S4 Table). The modulatory activity of these drugs is highly dependent on the protein structure, as evidenced by the fact that missense mutations in *SF3B1* confer resistance to the drugs [24,25]. By modifying four amino acids, we reproduced the human HEAT repeat 15 in *C. elegans* SFTB-1. As a result, worms became sensitive to pladienolide B and herboxidiene but maintained resistance to sudemycin D6. Further humanization of SFTB-1 and other SF3B complex components would

likely increase the sensitivity to these drugs and expand the number of known splicing modulators effective in *C. elegans*. As an example, while PHF5A is highly conserved in *C. elegans*, the Y36 position, which confers resistance to PB, HB, and sudemycin D6 when mutated [25], corresponds to a histidine. Thus, the H36Y substitution in *C. elegans* might increase sensitivity to splicing modulators.

It is worth mentioning that this is the first report of an invertebrate animal model sensitive to PB and HB. Previously, splicing inhibition by PB *in vivo* was reported in yeast strains that were edited to humanize some Hsh155 HEAT repeats and mutated to block PB efflux [26]. As an added value to the *C. elegans* toolkit, we have generated a strain that presents WT characteristics but is sensitive to splicing modulators such as PB and HB, which we believe has great potential for splicing-related studies in this nematode.

SF3B1 mutations are mutually exclusive with other splicing factor mutations present in tumors. *U2AF1* and *SRSF2* missense mutations are reported in 11% and 12–15% of MDS, respectively [19], and very rarely overlap with *SF3B1* mutations, which are present in 17–28% of MDS patients [37,39]. Hence, cells may not handle all the AS defects produced by the combination of mutations in any of these three proteins. To further validate our model as a tool for identifying vulnerabilities in cells with compromised SF3B1 function, we mimicked *U2AF1*^{S34F} and *SRSF2*^{P95H} mutations in *C. elegans* and detected a functional interaction between both mutants and the *sftb-1*[Q552P, R643C, K718E] allele. Additionally, *C. elegans* rsp-4 and *uaf-2* missense mutations could be used for RNA-seq analyses and RNAi screens to study the consequences and vulnerabilities of these mutations, respectively.

We have demonstrated that *C. elegans* can be a valuable model to explore vulnerabilities of cells with mutations in *SF3B1*, although the molecular consequences of *sftb-1* mutations were different from the ones observed in other organisms. Replacement of several HEAT repeats in yeast Hsh155 with their human counterparts is functional and supports splicing [60]. Similarly, partial or total replacement of *sftb-1* with its human ortholog could result in a more accurate model. To date, only one human gene has been functionally transplanted to *C. elegans* [61], in part due to low efficiencies of inserting long DNA fragments by CRISPR. Such technical difficulties can now be bypassed by the use of Nested CRISPR [62]. Thus, humanized *sftb-1* nematodes obtained by this method would be very efficient as *in vivo* models to further understand the impact of *SF3B1* mutations in cancer and as a platform for large-scale genetic and pharmacological screens.

Materials and methods

Caenorhabditis elegans strains

C. elegans strains were maintained using standard methods [63,64] at the specified temperature. Before conducting the experiments, strains were grown for at least two generations at the experimental temperature. Unless otherwise specified, we used Bristol N2 as the WT strain. Mutant strains with single *sftb-1* missense mutations (CER218, CER220, CER236, and CER238) were outcrossed x2. WT siblings resulting from the second outcross were given specific strain names (CER217, CER221, CER237, and CER239, respectively) and kept for closer comparison. CER351, CER442, and CER450 strains were outcrossed x2. In order to maintain the *cer6* deletion, heterozygous mutants were crossed with the EG7893: *oxTi615 unc-119(ed3) III* strain so that a red fluorescent marker was inserted 0.15 cM away from the *sftb-1* WT locus. To study the lasting of maternally inherited SFTB-1, we made the compound heterozygous strain CER505: *sftb-1(cer6/cer114[mCherry::sftb-1]) III*. All the strains used in this study are listed in S5 Table.

Protein alignments

H. sapiens SF3B1 (Uniprot: O75533) and *C. elegans* SFTB-1 (Uniprot: G5EEQ8) protein sequences were aligned using the T-Coffee algorithm [65] with default settings. <u>S1 Fig</u> was generated with Jalview [66].

CRISPR/Cas9 mutant and reporter strains

Specific guide RNAs were designed using both *Benchling* (www.benchling.com) and *CCTop* [67] online tools. All CRISPR/Cas9 mutant and reporter strains were obtained following a co-CRISPR strategy [68] using *dpy-10* as a marker to enrich for genome-editing events. In all cases, mixes were injected into gonads of young adult P_0 hermaphrodites using the Xeno-Works Microinjection System and following standard *C. elegans* microinjection techniques. F_1 progeny was screened by PCR using specific primers and F_2 homozygotes were confirmed by Sanger sequencing. As we optimized the protocol for CRISPR/Cas9 genome editing in our laboratory while preparing this manuscript, different reagents and conditions were used to produce the distinct mutants. The *sftb-1(cer6)* deletion mutant allele was obtained by nonspecific double-strand break repair on our first attempt to generate the K718E mutation. Only in this experiment, guide RNAs and Cas9 were expressed from plasmids that were purified with the NucleoBond Xtra Midi Kit (Macherey-Nagel). In the remaining mixes, ribonucleoprotein complexes containing crRNA, tracrRNA, and Cas9 were annealed at 37°C for 15 minutes prior to injection. All the reagents and injection mixes used in this study are listed in S6 Table.

C. elegans microscopy

Worms were mounted on 2% agar pads with 10 mM tetramisole hydrochloride and imaged using a ZEISS Axio Observer Z1 inverted fluorescence microscope. Images were processed using ZEISS ZEN 2012 (blue edition) software and Fiji.

Gonad dissection and DAPI staining

Worms expressing mCherry::SFTB-1 were anesthetized with 0.33 mM tetramisole hydrochloride and their heads cut using a pair of 25-gauge needles to extrude gonad arms. Released gonads were fixed in 4% paraformaldehyde in PBS for 20–30 minutes and washed with PBS + 0.1% Tween 20 for 10 minutes. After performing 3 washes, gonads were transferred to a microscope slide and mounted using 10 μ l DAPI Fluoromount-G (SouthernBiotech). Images were acquired the same day to avoid mCherry signal loss.

sftb-1 deletion mutant larval arrest quantification

Synchronized L1-arrested F_2 worms from heterozygous *sftb-1(cer6)/oxTi615* P_0 s were seeded onto NGM plates with OP50 and imaged after 48 hours of growth at 20°C using a ZEISS Axio Observer Z1 inverted fluorescence microscope at 10X magnification. Body length was measured manually by drawing the midline of the worm using the freehand line tool in Fiji.

Phenotypic characterization

Similar experiments were conducted to characterize *sftb-1*[Q552P, R643C, K718E], *rsp-4* [P100H], and *uaf-2*[S42F] mutant phenotypes. Briefly, synchronized L1-arrested worms of the specified genotype were first seeded onto NGM plates with OP50 and, after a few hours to allow recovery from the L1 arrest, 60 animals per genotype were separated individually to 12-well NGM plates seeded with OP50. Plates were incubated at 20°C (Figs 4 and S3) or 25°C (Fig 1F–1H). The developmental stage of individual worms was assessed visually every 24

hours under a stereomicroscope based on animal size and germline morphology. Phenotypes were scored at day 4. In order to determine brood sizes of *sftb-1*[Q552P, R643C, K718E] and WT animals, a proportion of worms were transferred into new 35-mm NGM plates at day 4 and subsequently moved to fresh plates every 12–24 h until egg laying ceased. Progeny were counted 36–48 h after passages.

RNA isolation

To study the transcriptional consequences of the K718E mutation, we used two homozygous mutant lines (CER218 and CER220) derived from two independent CRISPR/Cas9 events (*cer3* and *cer7* alleles, respectively), and the corresponding WT siblings (CER217 and CER221). Synchronized L4 animals were collected in M9 after growing for 41 hours at 20°C and total RNA was isolated using TRI Reagent (MRC, Inc.) and the PureLink RNA Mini Kit (Ambion), following the manufacturer's instructions.

sftb-1(cer6) homozygous larvae were obtained by automatically sorting non-red animals from a synchronized CER191 population grown at 20°C for 23 hours, using the COPAS Biosorter system (Union Biometrica). We conducted two biological replicates with an average of 1.6% non-red recombinants and 5.25% red contaminants in the samples. The same size and fluorescence window were used to sort two N2 WT populations collected under the same experimental conditions. Two CER276 biological replicates were collected independently without sorting, after growing for 23 hours at 20°C. Total RNA from the six samples was extracted using TRI Reagent, followed by aqueous and organic phase separation using 1-Bromo-3-chloropropane (Sigma-Aldrich), and RNA was precipitated with 2-propanol. After treatment with TURBO DNase (Thermo Fisher Scientific), RNA was purified following a phenol/chloroform extraction protocol.

In all cases, total RNA was quantified by Qubit RNA BR Assay kit (Thermo Fisher Scientific) and the integrity was checked by using the Agilent RNA 6000 Nano Kit (Agilent).

RNA sequencing

Four RNA-seq libraries (from CER217, CER218, CER220, and CER221 strains) were prepared with the KAPA Stranded mRNA-Seq Illumina Platforms Kit (Roche-Kapa Biosystems) following the manufacturer's recommendations. Briefly, 500 ng of total RNA was used as the input material, the poly-A fraction was enriched with oligo-dT magnetic beads and the mRNA was fragmented. Strand specificity was achieved during the second strand synthesis performed in the presence of dUTP instead of dTTP. The blunt-ended double stranded cDNA was 3'adeny-lated and Illumina indexed adapters from TruSeq[™] Stranded Total RNA Sample Preparation Kit with Ribo-Zero Gold (Illumina) were ligated.

The remaining six samples (two biological replicates from N2, CER276, and homozygous *sftb-1(cer6)* larvae) were prepared using the TruSeq[™] Stranded Total RNA kit protocol (Illumina). Briefly, rRNA was depleted from 500 ng of total RNA using the Ribo-Zero Gold rRNA Removal Kit and fragmented by divalent cations, with a major peak at 160 nt. Following the fragmentation, first and second strand synthesis was performed, also ensuring the library directionality. The cDNA was adenylated and ligated to xGen Dual Index UMI Adapters (IDT) for paired-end sequencing.

Both types of library ligation products were enriched with 15 PCR cycles and the final library was validated on an Agilent 2100 Bioanalyzer with the Agilent DNA 1000 Kit (Agilent).

The libraries were sequenced on HiSeq 2000 (Illumina) in paired-end mode with a read length of 2x76 bp using the HiSeq SBS Kit V4 50 cycle kit in a fraction of a sequencing v4 flow cell lane (HiSeq PE Cluster Kit V4—cBot). Image analysis, base calling, and quality scoring of

the run were processed using the manufacturer's software Real Time Analysis (RTA 1.18.66.3) and followed by generation of FASTQ sequence files.

Bioinformatic analysis

RNA-seq reads were mapped against the *C. elegans* reference assembly (WBcel235.87) with STAR [69]. Genes and transcripts were quantified with RSEM [70]. Differential expression analysis was performed with DESeq2 R package [71], and detection of differential alternative splicing events was done with rMATS [33].

RNAi screen

Information about the selected splicing genes was collected from SpliceosomeDB [72] or from the literature. RNAi clones were obtained from the ORFeome library [73] or the Ahringer library [28] and clone insert size was validated by PCR. For screening, we used 24-well plates containing NGM supplemented with 50 μ g/ml ampicillin, 12.5 μ g/ml tetracycline and 3 mM IPTG. Each well was seeded with 80 μ l of bacterial culture and dsRNA expression was induced overnight at room temperature. WT and mutant worms were tested in duplicates. Between 10 and 20 worms synchronized at the L1 stage were seeded in each well, and phenotypes were scored visually at the adult stage (72–96 h post-seeding at 25°C). Bacteria expressing dsRNA against *gfp* were used as negative control. Screens in *sftb-1*[K718E] and *sftb-1*[R643C] backgrounds were performed twice, at two different times, with different batches of plates and RNAi cultures, and scored by two different persons. Several candidates were selected for further validation, but the existence of false negatives in the screen is possible due to the noisy nature of RNAi for essential genes in small plates.

Interaction between teg-4 and sftb-1 mutations

Worms of the indicated genotype were singled onto individual 35-mm NGM plates with OP50 at the L4 stage and transferred to new plates every 12 hours until all eggs were laid. Numbers of dead embryos and hatched larvae were scored 24 and 48 hours after passages, respectively. Since the double mutant *sftb-1*[K718E]; *teg-4*(*oz210*) could not be maintained in homozygosis, F_1 worms from *sftb-1*[K718E]; *teg-4*(*oz210*/+) P_0 worms, including homozygous and heterozygous animals for *teg-4*(*oz210*), were picked at the L4 stage and their progeny scored. These F_1 s were genotyped after conducting the experiment to include only *sftb-1*[K718E] and *sftb-1*[K718E]; *teg-4*(*oz210*) data.

Validation of candidates from the RNAi screen

In all cases, we used 55-mm or 35-mm NGM plates supplemented with 50 μ g/mL ampicillin, 12.5 μ g/mL tetracycline and 3 mM IPTG, seeded with specific RNAi bacterial clones. RNAi targeting *gfp* was used as negative control and experiments were performed at 25°C.

In Fig 3C and 3D and S6B Fig, synchronized L1-arrested worms of the indicated genotypes were seeded onto 55-mm plates, and imaged directly on the plates after 48 hours of growth using a stereomicroscope. Body length was determined automatically with WormSizer [74]. In Fig 3E–3G, synchronized L1-arrested animals were seeded onto 55-mm plates and separated into individual 35-mm plates at the L4 stage. Worms were transferred every 24 hours and the presence of dead embryos was determined 24 hours after passages. In S6A Fig, synchronized L1-arrested animals were seeded onto 55-mm plates at the L4 stage. Worms were transferred every 24 hours and the presence of dead embryos was determined 24 hours after passages. In S6A Fig, synchronized L1-arrested animals were seeded onto 55-mm plates at the L4 stage, and the fertility phenotypes assessed in 2-day old adults.

Number of germ cells assay

WT or *sftb-1(cer6)* homozygous worms were collected at the indicated time points, fixed using Carnoy's solution (60% absolute ethanol, 30% chloroform, 10% acetic acid), and washed three times in PBS + 0.1% Tween 20. Whole worms were transferred to a microscope slide and stained with DAPI Fluoromount-G in order to visualize and count germ cell nuclei.

Maternal product in sftb-1(cer6) homozygotes

In order to monitor the presence of mCherry::SFTB-1 maternal product, sftb-1(cer114) or sftb-1(cer6/cer114) animals were separated blindly and allowed to lay eggs for 2, 9, and 12 hours in three separate plates in order to semi-synchronize the F₁ populations. Thus, only a fraction (23 hours) of the whole egg-laying period was covered. Red expression levels were assessed at different moments in order to include all the stages represented, which were designated visually based on anatomic features. sftb-1(cer6/cer114) animals were distinguished from sftb-1(cer114) animals by the presence of non-red arrested larvae in the next generation.

Semi-quantitative reverse-transcriptase PCR

N2, CER442, and CER450 animals grown at 20°C were collected at the L2 (24 hours after seeding L1-arrested worms) and adult stages. Total RNA was extracted using TRI Reagent, followed by aqueous and organic phase separation using 1-Bromo-3-chloropropane (Sigma-Aldrich), and RNA was precipitated with 2-propanol. After treatment with DNase I (Thermo Fisher Scientific), cDNA was synthetized with the H Minus First Strand cDNA Synthesis Kit (Thermo Fisher Scientific) following the manufacturer's instructions. Primer sequences are available upon request.

Spliceosome inhibitors experiments

Experiments in liquid media were conducted in 96-well plates. Experiments in S4 Table were conducted as follows: 25 μ l of dead OP50 resuspended in a mix of freshly prepared S Medium [64], 4 μ g/ml cholesterol, 250 μ g/ml streptomycin, and 62.5 μ g/ml tetracycline were added to each well. Subsequently, 25 μ l 2.4X drug and 10 μ l L1-arrested worms at a density of 1–2 worms/ μ l were added to each well. Each condition was tested in duplicate in the same 96-well plate. Plates were incubated in a humid chamber with gentle agitation at room temperature for the indicated time period. 25 μ l water (experiments 1–3) or 0.5% DMSO (Sigma-Aldrich) (experiments 4 and 6) were used as negative controls. In experiment 5, 50 μ M sudemycin D6 or 0.5% DMSO were injected into the gonads or intestine of young adult hermaphrodites using standard microinjection techniques, and injected animals were recovered in individual 35-mm NGM plates with OP50 and incubated at 20°C.

The protocol for liquid experiments in Fig 5C and 5D was slightly modified. In brief, 64.5 μ l S Medium supplemented with 5 μ g/ml cholesterol, 50 μ g/ml streptomycin, and 50 μ g/ml ampicillin were added to each well. This supplemented S Medium was used to resuspend a pellet of dead OP50 to an optical density of 1.1–1.2, and 25 μ l of bacteria were dispensed to each well. Subsequently, 10 μ l L1-arrested worms at a density of 4–6 worms/ μ l, and 0.5 μ l 200X drug were added to each well. Each condition was tested in triplicate in the same 96-well plate. Plates were incubated in a humid chamber at 25°C.

Statistical analyses

In all figure legends, 'N' denotes the number of independent replicate experiments performed, while 'n' indicates the total number of animals analyzed in each condition (different genotypes

are separated by commas). Statistical analyses were performed in GraphPad Prism 6 and R. Statistical tests used are reported in the figure legends. Figs <u>1A</u>, <u>1D</u>, <u>1F–1H</u>, <u>2B–2D</u>, <u>3B–3G</u>, <u>4</u>, <u>5C and 5D</u>, and some panels in <u>S2</u>, <u>S3</u>, <u>S4</u>, <u>S5</u> and <u>S6</u> Figs were generated by using the ggplot2 R package.

Supporting information

S1 Fig. Most of the SF3B1 mutated residues are conserved in the *C. elegans* **ortholog SFTB-1.** Protein sequence alignment of human SF3B1 (Hs) and worm SFTB-1 (Ce). The yellow histogram and the numerical index represent the degree of conservation (* corresponds to the maximum degree of conservation). U2AF ligand motifs (ULMs), p14-interacting region and HEAT domains are indicated in boxes. A cross symbol (X) below the conservation score indicates positions where missense mutations have been reported in human cancers (COSMIC database) at different frequencies. Residues that are relevant for this study are highlighted in green.

(TIF)

S2 Fig. cer6 homozygotes inherit SFTB-1 maternal product. (A) Schematic representation of the sftb-1 locus. Gray boxes indicate exons, connecting lines indicate introns. Left and right orange boxes represent predicted 5' and 3' untranslated regions, respectively. In pink, 29 nucleotides that are deleted in the cer6 allele. Predicted amino acids resulting from translating the WT or cer6 alleles are shown below the DNA sequence. The frameshift caused by the deletion introduces a premature stop codon in the cer6 transcript (colored in red). Scale bar, 100 base pairs (bp). (B) Germ cell number of WT (N2 strain) or homozygous sftb-1(cer6) synchronized larvae grown for 14, 23, and 40 hours at 20°C. In WT worms, differentiating spermatocytes were not included at the last time point, and the number of germ cells was counted in one gonad arm only. Black and gray lines connect the median germ cell number at each time point in WT and *sftb-1(cer6)* worms, respectively. The dotted line represents the four germ cells that are normally present in recently hatched L1 larvae; these cells will resume proliferation during the L1 stage. Between 2 and 10 animals per condition were scored (N = 1). (C) Schematic representation of the compound heterozygous strain CER505: sftb-1(cer6)/sftb-1 (cer114[mCherry::sftb-1]) III. (D) Total number of progeny laid by heterozygous (cer6/+) or WT (+/+) worms from a CER505 population (n = 7, 9; N = 1). The number of F_1 larvae expressing mCherry::SFTB-1 was not significantly different between both groups (n.s.), while cer6/+ animals laid a significantly higher number of dead embryos which lacked red fluorescence (Mann-Whitney's test; * p<0.05). Heterozygotes also segregated a number of non-red larvae that were arrested and were not observed in WT plates, indicating that they were cer6/ *cer6* animals. '+' denotes the *cer114[mCherry::sftb-1]* allele. Dots represent measures in individual worms, overlaid to Tukey-style boxplots. (E) Mean percentage of non-red F₁ arrested larvae (% Lva), F1 dead embryos (% Emb), and mean brood size in worms of the indicated genotypes, from data represented in panel D. (F) mCherry::SFTB-1 expression in F₁ progeny from wild-type (top, n = 3) or heterozygous (bottom, n = 9) animals from a CER505 population (N = 1). At each stage, vertical lines correspond to individual worms, and solid red, open red, and black dots represent the total number of red, dim red, or non-red progeny observed, respectively. Representative DIC and fluorescence microscopy images are shown below the graph (scale bar, 25 µm). The mean percentage of dim red and non-red progeny laid by heterozygotes is indicated in late stages and is not specified in early stages as it was 0%. (TIF)

S3 Fig. Molecular design of sftb-1 cancer-related mutations and sftb-1 [Q552P, R643C, K718E] phenotypes at 20°C. (A) Schematic representation of the *sftb-1* locus, as in S2 Fig. The position of the triplets encoding the three mutated residues in this study (Q552, R643 and K718) is indicated. Scale bar, 100 bp. (B) Molecular details of the three sftb-1 missense mutations generated by CRISPR/Cas9. In WT alleles, the crRNA sequence used is indicated by orange nucleotides, the protospacer-adjacent motif (PAM) is underlined and the Cas9 cut site is indicated with a black arrowhead. In mutant alleles, synonymous mutations introduced to prevent partial recombination events and to improve primer specificity for mutant allele detection by PCR are indicated in blue. Mutated codons and the corresponding amino acids are colored in pink. (C) Bars represent mean incidence of different phenotypes observed in sftb-1 [Q552P, R643C, K718E] worms at 20°C, while dots represent percentages observed in each replicate (n = 116, 109; N = 2). (D) Brood size of WT or *sftb-1*[Q552P, R643C, K718E] worms at 20° C (n = 33, 46; N = 2). Dots represent values for each individual animal, overlaid to Tukey-style boxplots. (E) sftb-1[Q552P, R643C, K718E] animals present a mild developmental delay at 20 $^{\circ}$ C. Dot sizes represent the proportion of the population at each stage (n = 116, 109; N = 2). Statistics: (C), Fisher's exact test (D), Student's t-test (unpaired, two-tailed). n.s., no significant difference, **** p<0.0001. (TIF)

S4 Fig. Pathway enrichment analysis of genes with differentially expressed transcripts in different *sftb-1* mutant strains. Significantly enriched functional terms in the distinct datasets are listed (padj<0.01). Bars represent the adjusted enrichment p-values in negative log_{10} scale, color-coded by mutant strain. Solid and clear bars denote enriched terms in upregulated and downregulated genes, respectively. The number of genes with differentially expressed transcripts belonging to each category is shown. The analysis was performed with the g:GOSt tool in g:Profiler, and only biological pathways from KEGG and WikiPathways databases are shown.

(TIF)

S5 Fig. Common AS events deregulated by *sftb-1* **alleles.** (A) Summary of the overlapping of genes with AS changes (left) and the overlapping of AS events (right) between groups. (B) Venn diagram displaying the overlapping AS events between the three datasets. The total number of significant events with inclusion level difference > 0.1 in each dataset is indicated. (C) Plot showing the inclusion level difference (InclLevelDifference) of AS events that were deregulated in at least two datasets ('common AS events'), color-coded according to the dataset. Different genes are represented along the x-axis, and events are separated by event type. Some genes (*acdh-2, fbn-1, Y69H2.3, unc-89, swp-1,* and *ate-1*) shared more than one event in different datasets.

(TIF)

S6 Fig. *sftb-1(cer6)* heterozygotes are sensitive to RNAi of five interactors. (A) *mog-2* (*RNAi*), *smr-1(RNAi*), and *teg-4(RNAi*) cause a significant mild increase in sterility in *sftb-1* (*cer6*) heterozygotes compared to WT at 25°C ($n \ge 56$; N = 2; * p < 0.05, ** p < 0.01, **** p < 0.0001; Fisher's exact test). Red dots indicate percent sterility observed in each replicate. Gray bars, P₀-treated worms giving rise to <5 F₁ larvae and some dead embryos ('dead progeny' category); black bars, P₀-treated worms that laid neither larvae nor dead embryos ('no progeny' category). (B) *sftb-1(RNAi*) and *uaf-2(RNAi*) induce an earlier larval arrest in *cer6/* + heterozygous worms. Dots represent individual worm lengths after 48 h of RNAi treatment at 25°C, overlaid to Tukey-style boxplots ($n \ge 424$; N = 2; **** p < 0.0001; Kruskal-Wallis test with Dunn's multiple comparison test). The genotype of the strain used for these experiments

was CER190: *sftb-1(cer6)/dpy-17(e164) unc-79(e1068) III*. (TIF)

S7 Fig. Effect of *uaf-2*[S42F] and *rsp-4*[P100H] mutations on a subset of AS events. (A) cDNA samples from WT (N2), *uaf-2*[S42F], and *rsp-4*[P100H] worms at L2 and adult stage were used to analyze the presence of alternative isoforms by semiquantitative (sq) RT-PCR (N = 1). The alternative exons are numbered and shaded in gray in the schematic representation on the right, together with upstream and downstream exons where the forward and reverse primers annealed, respectively. The three distinct isoforms in *unc-32* E4 are indicated with red lines on the left. *act-1* was used as an endogenous control. (B) The 3'SS sequence of the alternative exons is indicated, as well as the CCNG and GGNG motifs found in alternative exons using FIMO with a p-value < 0.01. (C) Technical replicate of *tos-1* E1-E2 and *tos-1* E1-E4 sqRT-PCRs to verify that *uaf-2*[S42F] promotes intron 1 retention (slowest migrating band in both gels) and exon 3 skipping (fastest migrating band in bottom gel), both indicated with blue boxes. 'E' indicates exon, while 'I' indicates intron. (TIF)

S1 Table. Alternative splicing analysis of *sftb-1* mutants by rMATS. (XLSX)

S2 Table. Differential expression analysis of *sftb-1* mutants by DESeq2. (XLSX)

S3 Table. Information and phenotypes of the RNAi splicing screen. (XLSX)

S4 Table. First assays with spliceosome inhibitors. (XLSX)

S5 Table. *C. elegans* strains used in this study. (XLSX)

S6 Table. CRISPR reagents and injection mixes. (XLSX)

Acknowledgments

We acknowledge the members of the Cerón Laboratory for helpful discussions and comments on the manuscript. We also thank Alan Zahler and Sol Katzman for their advice in transcriptomic analyses. The *teg-4(oz210)* strain was kindly provided by Dave Hansen. Pladienolide B and sudemycin D6 were gifts from Juan Valcárcel. We thank CERCA Program / Generalitat de Catalunya for their institutional support.

Author Contributions

Conceptualization: Julián Cerón.

Formal analysis: Xènia Serrat, Anna Esteve-Codina.

Funding acquisition: Julián Cerón.

Investigation: Xènia Serrat, Dmytro Kukhtar, Eric Cornes, Julián Cerón.

Methodology: Xènia Serrat, Dmytro Kukhtar, Eric Cornes, Helena Benlloch, Germano Cecere.

Supervision: Julián Cerón.

Writing – original draft: Xènia Serrat, Julián Cerón.

Writing – review & editing: Xènia Serrat, Julián Cerón.

References

- Urbanski LM, Leclair N, Anczuków O. Alternative-splicing defects in cancer: Splicing regulators and their downstream targets, guiding the way to novel cancer therapeutics. WIREs RNA. 2018; 9: e1476. https://doi.org/10.1002/wrna.1476 PMID: 29693319
- Yoshida K, Sanada M, Shiraishi Y, Nowak D, Nagata Y, Yamamoto R, et al. Frequent pathway mutations of splicing machinery in myelodysplasia. Nature. 2011; 478: 64–69. <u>https://doi.org/10.1038/ nature10496 PMID: 21909114</u>
- Papaemmanuil E, Cazzola M, Boultwood J, Malcovati L, Vyas P, Bowen D, et al. Somatic SF3B1 Mutation in Myelodysplasia with Ring Sideroblasts. N Engl J Med. 2011; 365: 1384–1395. https://doi.org/10. 1056/NEJMoa1103283 PMID: 21995386
- Malcovati L, Karimi M, Papaemmanuil E, Ambaglio I, Jädersten M, Jansson M, et al. SF3B1 mutation identifies a distinct subset of myelodysplastic syndrome with ring sideroblasts. Blood. 2015; 126: 233– 241. https://doi.org/10.1182/blood-2015-03-633537 PMID: 25957392
- Quesada V, Conde L, Villamor N, Ordóñez GR, Jares P, Bassaganyas L, et al. Exome sequencing identifies recurrent mutations of the splicing factor SF3B1 gene in chronic lymphocytic leukemia. Nat Genet. 2012; 44: 47–52. https://doi.org/10.1038/ng.1032 PMID: 22158541
- te Raa G, Derks I, Navrkalova V, Skowronska A, Moerland P, van Laar J, et al. The impact of SF3B1 mutations in CLL on the DNA-damage response. Leukemia. 2015; 29: 1133–1142. <u>https://doi.org/10. 1038/leu.2014.318 PMID: 25371178</u>
- Wang L, Lawrence MS, Wan Y, Stojanov P, Sougnez C, Stevenson K, et al. SF3B1 and Other Novel Cancer Genes in Chronic Lymphocytic Leukemia. N Engl J Med. 2011; 365: 2497–2506. https://doi.org/ 10.1056/NEJMoa1109016 PMID: 22150006
- Furney SJ, Pedersen M, Gentien D, Dumont AG, Rapinat A, Desjardins L, et al. SF3B1 mutations are associated with alternative splicing in uveal melanoma. Cancer Discov. 2013; 3: 1122–1129. https://doi. org/10.1158/2159-8290.CD-13-0330 PMID: 23861464
- Harbour J, Roberson E, Anbunathan H, Onken M, Worley L, Bowcock M. Recurrent mutations at codon 625 of the splicing factor *SF3B1* in uveal melanoma. Nat Genet. 2013; 45: 133–135. <u>https://doi.org/10. 1038/ng.2523 PMID: 23313955</u>
- Martin M, Maßhöfer L, Temming P, Rahmann S, Metz C, Bornfeld N, et al. Exome sequencing identifies recurrent somatic mutations in *EIF1AX* and *SF3B1* in uveal melanoma with disomy 3. Nat Genet. 2013; 45: 933–936. https://doi.org/10.1038/ng.2674 PMID: 23793026
- Biankin A V., Waddell N, Kassahn KS, Gingras MC, Muthuswamy LB, Johns AL, et al. Pancreatic cancer genomes reveal aberrations in axon guidance pathway genes. Nature. 2012; 491: 399–405. https:// doi.org/10.1038/nature11547 PMID: 23103869
- Maguire SL, Leonidou A, Wai P, Marchiò C, Ng CK, Sapino A, et al. SF3B1 mutations constitute a novel therapeutic target in breast cancer. J Pathol. 2015; 235: 571–580. https://doi.org/10.1002/path.4483 PMID: 25424858
- Darman RB, Seiler M, Agrawal AA, Lim KH, Peng S, Aird D, et al. Cancer-Associated SF3B1 Hotspot Mutations Induce Cryptic 3' Splice Site Selection through Use of a Different Branch Point. Cell Rep. 2015; 13: 1033–1045. https://doi.org/10.1016/j.celrep.2015.09.053 PMID: 26565915
- DeBoever C, Ghia EM, Shepard PJ, Rassenti L, Barrett CL, Jepsen K, et al. Transcriptome sequencing reveals potential mechanism of cryptic 3' splice site selection in SF3B1-mutated cancers. PLoS Comput Biol. 2015; 11: e1004105. https://doi.org/10.1371/journal.pcbi.1004105 PMID: 25768983
- Liberante FG, Lappin K, Barros EM, Vohhodina J, Grebien F, Savage KI, et al. Altered splicing and cytoplasmic levels of tRNA synthetases in SF3B1-mutant myelodysplastic syndromes as a therapeutic vulnerability. Sci Rep. 2019; 9: 2678. https://doi.org/10.1038/s41598-019-39591-7 PMID: 30804405
- Alsafadi S, Houy A, Battistella A, Popova T, Wassef M, Henry E, et al. Cancer-associated SF3B1 mutations affect alternative splicing by promoting alternative branchpoint usage. Nat Commun. 2016; 7:10615. https://doi.org/10.1038/ncomms10615 PMID: 26842708
- Gupta AK, Murthy T, Paul K V, Ramirez O, Fisher JB, Rao S, et al. Degenerate minigene library analysis enables identification of altered branch point utilization by mutant splicing factor 3B1 (SF3B1). Nucleic Acids Res. 2019; 47: 970–980. https://doi.org/10.1093/nar/gky1161 PMID: 30462273

- Shiozawa Y, Malcovati L, Gallì A, Sato-Otsubo A, Kataoka K, Sato Y, et al. Aberrant splicing and defective mRNA production induced by somatic spliceosome mutations in myelodysplasia. Nat Commun. 2018; 9: 3649. https://doi.org/10.1038/s41467-018-06063-x PMID: 30194306
- Xu JJ, Smeets MF, Tan SY, Wall M, Purton LE, Walkley CR. Modeling human RNA spliceosome mutations in the mouse: not all mice were created equal. Exp Hematol. 2019; 70: 10–23. <u>https://doi.org/10. 1016/j.exphem.2018.11.001 PMID: 30408513</u>
- Bonnal S, Vigevani L, Valcárcel J. The spliceosome as a target of novel antitumour drugs. Nat Rev Drug Discov. 2012; 11: 847–59. https://doi.org/10.1038/nrd3823 PMID: 23123942
- Effenberger KA, Urabe VK, Jurica MS. Modulating splicing with small molecular inhibitors of the spliceosome. WIREs RNA. 2017; 8: e1381. https://doi.org/10.1002/wrna.1381 PMID: 27440103
- DeNicola AB, Tang Y. Therapeutic approaches to treat human spliceosomal diseases. Curr Opin Biotechnol. 2019; 60: 72–81. https://doi.org/10.1016/j.copbio.2019.01.003 PMID: 30772756
- Seiler M, Yoshimi A, Darman R, Chan B, Keaney G, Thomas M, et al. H3B-8800, an orally available small-molecule splicing modulator, induces lethality in spliceosome-mutant cancers. Nat Med. 2018; 24: 497–504. https://doi.org/10.1038/nm.4493 PMID: 29457796
- 24. Yokoi A, Kotake Y, Takahashi K, Kadowaki T, Matsumoto Y, Minoshima Y, et al. Biological validation that SF3b is a target of the antitumor macrolide pladienolide. FEBS J. 2011; 278: 4870–4880. <u>https://doi.org/10.1111/j.1742-4658.2011.08387.x PMID: 21981285</u>
- 25. Teng T, Tsai JH, Puyang X, Seiler M, Peng S, Prajapati S, et al. Splicing modulators act at the branch point adenosine binding pocket defined by the PHF5A-SF3b complex. Nat Commun. 2017; 8: 15522. https://doi.org/10.1038/ncomms15522 PMID: 28541300
- Hansen SR, Nikolai BJ, Spreacker PJ, Carrocci TJ, Hoskins AA. Chemical Inhibition of Pre-mRNA Splicing in Living Saccharomyces cerevisiae. Cell Chem Biol. 2019; 26: 443–448. https://doi.org/10. 1016/j.chembiol.2018.11.008 PMID: 30639260
- Loerch S, Leach JR, Horner SW, Maji D, Jenkins JL, Pulvino M, et al. The pre-mRNA splicing and transcription factor Tat-SF1 is a functional partner of the spliceosome SF3b1 subunit via a U2AF homology motif interface. J Biol Chem. 2018; 294: 2892–2902. <u>https://doi.org/10.1074/jbc.RA118.006764</u> PMID: 30567737
- Kamath RS, Fraser AG, Dong Y, Poulin G, Durbin R, Gotta M, et al. Systematic functional analysis of the *Caenorhabditis elegans* genome using RNAi. Nature. 2003; 421: 231–237. <u>https://doi.org/10.1038/</u> nature01278 PMID: 12529635
- Cretu C, Schmitzová J, Ponce-Salvatierra A, Dybkov O, De Laurentiis EI, Sharma K, et al. Molecular Architecture of SF3b and Structural Consequences of Its Cancer-Related Mutations. Mol Cell. 2016; 64: 307–319. https://doi.org/10.1016/j.molcel.2016.08.036 PMID: 27720643
- Carrocci TJ, Zoerner DM, Paulson JC, Hoskins AA. SF3b1 mutations associated with myelodysplastic syndromes alter the fidelity of branchsite selection in yeast. Nucleic Acids Res. 2017; 45: 4837–4852. https://doi.org/10.1093/nar/gkw1349 PMID: 28062854
- Wang L, Brooks AN, Fan J, Wan Y, Gambe R, Li S, et al. Transcriptomic Characterization of SF3B1 Mutation Reveals Its Pleiotropic Effects in Chronic Lymphocytic Leukemia. Cancer Cell. 2016; 30: 750– 763. https://doi.org/10.1016/j.ccell.2016.10.005 PMID: 27818134
- Pellagatti A, Armstrong RN, Steeples V, Sharma E, Repapi E, Singh S, et al. Impact of spliceosome mutations on RNA splicing in myelodysplasia: dysregulated genes/pathways and clinical associations. Blood. 2018; 132: 1225–1240. https://doi.org/10.1182/blood-2018-04-843771 PMID: 29930011
- **33.** Shen S, Park JW, Lu Z, Lin L, Henry MD, Wu YN, et al. rMATS: Robust and flexible detection of differential alternative splicing from replicate RNA-Seq data. Proc Natl Acad Sci U S A. 2014; 111: E5593–601. https://doi.org/10.1073/pnas.1419161111 PMID: 25480548
- Kerins JA, Hanazawa M, Dorsett M, Schedl T. PRP-17 and the Pre-mRNA Splicing Pathway Are Preferentially Required for the Proliferation Versus Meiotic Development Decision and Germline Sex Determination in *Caenorhabditis elegans*. Dev Dyn. 2010; 239: 1555–1572. <u>https://doi.org/10.1002/dvdy.22274</u> PMID: 20419786
- 35. Mantina P, MacDonald L, Kulaga A, Zhao L, Hansen D. A mutation in *teg-4*, which encodes a protein homologous to the SAP130 pre-mRNA splicing factor, disrupts the balance between proliferation and differentiation in the *C. elegans* germ line. Mech Dev. 2009; 126: 417–429. <u>https://doi.org/10.1016/j.mod.2009.01.006</u> PMID: 19368799
- Imielinski M, Berger AH, Hammerman PS, Hernandez B, Pugh TJ, Hodis E, et al. Mapping the Hallmarks of Lung Adenocarcinoma with Massively Parallel Sequencing. Cell. 2012; 150: 1107–1120. https://doi.org/10.1016/j.cell.2012.08.029 PMID: 22980975

- Seiler M, Peng S, Agrawal AA, Palacino J, Teng T, Zhu P, et al. Somatic Mutational Landscape of Splicing Factor Genes and Their Functional Consequences across 33 Cancer Types. Cell Rep. 2018; 23: 282–296.e4. https://doi.org/10.1016/j.celrep.2018.01.088 PMID: 29617667
- Ilagan JO, Ramakrishnan A, Hayes B, Murphy ME, Zebari AS, Bradley P, et al. U2AF1 mutations alter splice site recognition in hematological malignancies. Genome Res. 2015; 25: 14–26. https://doi.org/10. 1101/gr.181016.114 PMID: 25267526
- Lee SC-W, North K, Kim E, Jang E, Obeng E, Lu SX, et al. Synthetic Lethal and Convergent Biological Effects of Cancer-Associated Spliceosomal Gene Mutations. Cancer Cell. 2018; 34: 225–241.e8. https://doi.org/10.1016/j.ccell.2018.07.003 PMID: 30107174
- Cretu C, Agrawal AA, Cook A, Will CL, Fekkes P, Smith PG, et al. Structural Basis of Splicing Modulation by Antitumor Macrolide Compounds. Mol Cell. 2018; 70: 265–273.e8. https://doi.org/10.1016/j. molcel.2018.03.011 PMID: 29656923
- Lagisetti C, Palacios G, Goronga T, Freeman B, Caufield W, Webb TR. Optimization of antitumor modulators of pre-mRNA splicing. J Med Chem. 2013; 56: 10033–10044. <u>https://doi.org/10.1021/jm401370h</u> PMID: 24325474
- 42. Finci LI, Zhang X, Huang X, Zhou Q, Tsai J, Teng T, et al. The cryo-EM structure of the SF3b spliceosome complex bound to a splicing modulator reveals a pre-mRNA substrate competitive mechanism of action. Genes Dev. 2018; 32: 309–320. https://doi.org/10.1101/gad.311043.117 PMID: 29491137
- Dickinson DJ, Goldstein B. CRISPR-Based Methods for Caenorhabditis elegans Genome Engineering. Genetics. 2016; 202: 885–901. https://doi.org/10.1534/genetics.115.182162 PMID: 26953268
- Nance J, Frøkjær-Jensen C. The Caenorhabditis elegans Transgenic Toolbox. Genetics. 2019; 212: 959–990. https://doi.org/10.1534/genetics.119.301506 PMID: 31405997
- 45. An M, Henion PD. The zebrafish sf3b1^{b460} mutant reveals differential requirements for the sf3b1 premRNA processing gene during neural crest development. Int J Dev Biol. 2012; 56: 223–237. <u>https://doi.org/10.1387/ijdb.113383ma PMID: 22562198</u>
- 46. Isono K, Mizutani-Koseki Y, Komori T, Schmidt-Zachmann MS, Koseki H. Mammalian Polycomb-mediated repression of *Hox* genes requires the essential spliceosomal protein Sf3b1. Genes Dev. 2005; 19: 536–541. https://doi.org/10.1101/gad.1284605 PMID: 15741318
- Zhou Q, Derti A, Ruddy D, Rakiec D, Kao I, Lira M, et al. A chemical genetics approach for the functional assessment of novel cancer genes. Cancer Res. 2015; 75: 1949–1958. <u>https://doi.org/10.1158/0008-5472.CAN-14-2930 PMID: 25788694</u>
- 48. Paolella BR, Gibson WJ, Urbanski LM, Alberta JA, Zack TI, Bandopadhayay P, et al. Copy-number and gene dependency analysis reveals partial copy loss of wild-type SF3B1 as a novel cancer vulnerability. Elife. 2017; 6: e23268. https://doi.org/10.7554/eLife.23268 PMID: 28177281
- 49. Dolatshad H, Pellagatti A, Fernandez-Mercado M, Yip B, Malcovati L, Attwood M, et al. Disruption of SF3B1 results in deregulated expression and splicing of key genes and pathways in myelodysplastic syndrome hematopoietic stem and progenitor cells. Leukemia. 2015; 29: 1092–1103. https://doi.org/10. 1038/leu.2014.331 PMID: 25428262
- Obeng EA, Chappell RJ, Seiler M, Chen MC, Campagna DR, Schmidt PJ, et al. Physiologic Expression of *Sf3b1*^{K700E} Causes Impaired Erythropoiesis, Aberrant Splicing, and Sensitivity to Therapeutic Spliceosome Modulation. Cancer Cell. 2016; 30: 404–417. <u>https://doi.org/10.1016/j.ccell.2016.08.006</u> PMID: 27622333
- Mupo A, Seiler M, Sathiaseelan V, Pance A, Yang Y, Agrawal A, et al. Hemopoietic-specific Sf3b1-K700E knock-in mice display the splicing defect seen in human MDS but develop anemia without ring sideroblasts. Leukemia. 2017; 31: 720–727. https://doi.org/10.1038/leu.2016.251 PMID: 27604819
- Yin S, Gambe RG, Sun J, Martinez AZ, Cartun ZJ, Regis FFD, et al. A Murine Model of Chronic Lymphocytic Leukemia Based on B Cell-Restricted Expression of *Sf3b1* Mutation and *Atm* Deletion. Cancer Cell. 2019; 35: 283–296.e5. https://doi.org/10.1016/j.ccell.2018.12.013 PMID: 30712845
- Dalton WB, Helmenstine E, Walsh N, Gondek LP, Kelkar DS, Read A, et al. Hotspot SF3B1 mutations induce metabolic reprogramming and vulnerability to serine deprivation. J Clin Invest. 2019; <u>https://doi.org/10.1172/JCl125022</u> PMID: 31393856
- 54. Morton JJ, Blumenthal T. RNA Processing in *C. elegans*. Methods in Cell Biology. Second Edi. 2011. pp. 187–217. https://doi.org/10.1016/B978-0-12-544172-8.00007–4
- Tourasse NJ, Millet JRM, Dupuy D. Quantitative RNA-seq meta-analysis of alternative exon usage in C. elegans. Genome Res. 2017; 27: 2120–2128. https://doi.org/10.1101/gr.224626.117 PMID: 29089372
- Pan Q, Shai O, Lee LJ, Frey BJ, Blencowe BJ. Deep surveying of alternative splicing complexity in the human transcriptome by high-throughput sequencing. Nat Genet. 2008; 40: 1413–1415. <u>https://doi.org/ 10.1038/ng.259</u> PMID: 18978789

- Ragle JM, Katzman S, Akers TF, Barberan-Soler S, Zahler AM. Coordinated tissue-specific regulation of adjacent alternative 3' splice sites in *C. elegans*. Genome Res. 2015; 25: 982–94. https://doi.org/10. 1101/gr.186783.114 PMID: 25922281
- Kfir N, Lev-Maor G, Glaich O, Alajem A, Datta A, Sze SK, et al. SF3B1 association with chromatin determines splicing outcomes. Cell Rep. 2015; 11: 618–29. https://doi.org/10.1016/j.celrep.2015.03.048
 PMID: 25892229
- Wu G, Fan L, Edmonson MN, Shaw T, Boggs K, Easton J, et al. Inhibition of SF3B1 by molecules targeting the spliceosome results in massive aberrant exon skipping. RNA. 2018; 24: 1056–1066. https://doi.org/10.1261/rna.065383.117 PMID: 29844105
- Carrocci TJ, Paulson JC, Hoskins AA. Functional analysis of Hsh155/SF3b1 interactions with the U2 snRNA/branch site duplex. RNA. 2018; 24: 1028–1040. https://doi.org/10.1261/rna.065664.118 PMID: 29752352
- McDiarmid TA, Au V, Loewen AD, Liang J, Mizumoto K, Moerman DG, et al. CRISPR-Cas9 human gene replacement and phenomic characterization in *Caenorhabditis elegans* to understand the functional conservation of human genes and decipher variants of uncertain significance. Dis Model Mech. 2018; 11: dmm036517. https://doi.org/10.1242/dmm.036517 PMID: 30361258
- Vicencio J, Martínez-Fernández C, Serrat X, Cerón J. Efficient Generation of Endogenous Fluorescent Reporters by Nested CRISPR in *Caenorhabditis elegans*. Genetics. 2019; 211: 1143–1154. https://doi. org/10.1534/genetics.119.301965 PMID: 30696716
- Porta-de-la-Riva M, Fontrodona L, Villanueva A, Cerón J. Basic Caenorhabditis elegans Methods: Synchronization and Observation. J Vis Exp. 2012; 64: e4019. https://doi.org/10.3791/4019 PMID: 22710399
- Stiernagle T. Maintenance of C. elegans. WormBook. 2006. pp. 1–11. <u>https://doi.org/10.1895/</u> wormbook.1.101.1 PMID: 18050451
- Notredame C, Higgins DG, Heringa J. T-coffee: A novel method for fast and accurate multiple sequence alignment. J Mol Biol. 2000; 302: 205–217. https://doi.org/10.1006/jmbi.2000.4042 PMID: 10964570
- Waterhouse AM, Procter JB, Martin DMA, Clamp M, Barton GJ. Jalview Version 2-A multiple sequence alignment editor and analysis workbench. Bioinformatics. 2009; 25: 1189–1191. https://doi.org/10. 1093/bioinformatics/btp033 PMID: 19151095
- Stemmer M, Thumberger T, Del Sol Keyer M, Wittbrodt J, Mateo JL. CCTop: An Intuitive, Flexible and Reliable CRISPR/Cas9 Target Prediction Tool. PLoS One. 2015; 10: e0124633. https://doi.org/10. 1371/journal.pone.0124633 PMID: 25909470
- Kim H, Ishidate T, Ghanta KS, Seth M, Conte D, Shirayama M, et al. A co-CRISPR strategy for efficient genome editing in *Caenorhabditis elegans*. Genetics. 2014; 197: 1069–80. <u>https://doi.org/10.1534/genetics.114.166389</u> PMID: 24879462
- Dobin A, Davis CA, Schlesinger F, Drenkow J, Zaleski C, Jha S, et al. STAR: ultrafast universal RNAseq aligner. Bioinformatics. 2013; 29: 15–21. <u>https://doi.org/10.1093/bioinformatics/bts635</u> PMID: 23104886
- 70. Li B, Dewey CN. RSEM: accurate transcript quantification from RNA-Seq data with or without a reference genome. BMC Bioinformatics. 2011; 12: 323. https://doi.org/10.1186/1471-2105-12-323 PMID: 21816040
- Love MI, Huber W, Anders S. Moderated estimation of fold change and dispersion for RNA-seq data with DESeq2. Genome Biol. 2014; 15: 550. https://doi.org/10.1186/s13059-014-0550-8 PMID: 25516281
- Cvitkovic I, Jurica MS. Spliceosome database: A tool for tracking components of the spliceosome. Nucleic Acids Res. 2013; 41: D132–D141. https://doi.org/10.1093/nar/gks999 PMID: 23118483
- 73. Rual J-F, Ceron J, Koreth J, Hao T, Nicot A-S, Hirozane-Kishikawa T, et al. Toward Improving *Caenor-habditis elegans* Phenome Mapping With an ORFeome-Based RNAi Library. Genome Res. 2004; 14: 2162–2168. https://doi.org/10.1101/gr.2505604 PMID: 15489339
- 74. Moore BT, Jordan JM, Baugh LR. WormSizer: High-throughput Analysis of Nematode Size and Shape. PLoS One. 2013; 8: e57142. https://doi.org/10.1371/journal.pone.0057142 PMID: 23451165
- 75. Forbes SA, Beare D, Boutselakis H, Bamford S, Bindal N, Tate J, et al. COSMIC: somatic cancer genetics at high-resolution. Nucleic Acids Res. 2017; 45: D777–D783. https://doi.org/10.1093/nar/gkw1121 PMID: 27899578